Author’s response to reviews

Title: Lifting the Veil on Disrespect and Abuse in Facility Based Child Birth Care: Findings from South West Nigeria.

Authors:

Macellina Ijadunola (yijadun@yahoo.com)
Ezeomu Olotu (ezeomuolotu@yahoo.com)
Olaitan Oyedun (nicholaite@yahoo.com)
Stanley Eferakeya (eferakeyastanley@yahoo.com)
Faith Ilesanmi (toytosin4u@yahoo.com)
Ayotomiwa Fagbemi (tmwfagbemi@yahoo.com)
Omowunmi Fasae (blutips@yahoo.com)

Version: 1 Date: 08 Jan 2019

Author’s response to reviews:

1. The minimum age for participants was 15 years and this has been included in the Results section, page 9 line 175

2. Verbal consent was taken from the legal guardian of participants who were less than 18 years and this has been included in the section on Ethical Approval, page 14, line 297.